Author's response to reviews

Title: Chronic Kidney Disease of uncertain aetiology; prevalence and causative factors in a developing country

Authors:

Nihal Jayatilake (secretary@health.gov.lk)
Palitha Maheepala (ddgphs@gmail.com)
Shanthi Mendis (Mendiss@who.int)
Firdosi Mehta (Mehtaf@who.int)

Version: 3 Date: 13 August 2013

Author's response to reviews:

Editor
BMC Nephrology
13.8.13

Dear Sir

MS: 4706172759459814  `Chronic Kidney Disease of uncertain aetiology ; prevalence and causative factors in a developing country`  

I wish to submit the manuscript entitled `Chronic Kidney Disease of uncertain aetiology ; prevalence and causative factors in a developing country` for consideration for publication in the BMC Nephrology.

Both reviewers were satisfied with the last revision submitted on the 24th July 2013, and had no further comments.

The manuscript has been copy edited. I wish to submit it as the corresponding author on behalf of the National CKDu project team.

Thank you for your consideration.

Yours sincerely

Dr Shanthi Mendis
MD FRCP FACC
Director a.i., Management of noncommunicable Diseases
World Health Organization, Geneva, Switzerland